\begin{thebibliography}{10}
\addcontentsline{toc}{chapter}{Bibliography}
\bibitem{Sinerva}
P. Sinerva, these proceedings;\\
CDF Collaboration (F. Abe et al.), \prl{74}{1995}{2626};\\
D0 Collaboration (S. Abachi et al.), \prl{74}{1995}{2632}.

\bibitem{bardin}
D. Bardin et al., in
                  Reports of the Working Group on Precision
                  Calculations for the $Z$-Resonance, CERN 95-03,
                  pg. 7.

\bibitem{HollikBrus}
W. Hollik, talk at Int. Europhysics Conf. on HEP, Brussels, 1995,
KA-TP-15-1995, hep-ph/9512232. 


\bibitem{Swartz}
M. Swartz , these proceedings, and references therein.
\bibitem{Kniehl}
For a recent compilation of theoretical results see B. Kniehl, in
                  Reports of the Working Group on Precision
                  Calculations for the $Z$-Resonance, CERN 95-03,
                  pg. 299.
\bibitem{Sirlin}
A. Sirlin, ibid. pg. 299.
\bibitem{degrassi}
G. Degrassi et al., ibid. pg. 163.

\bibitem{Hill}
W.A. Bardeen, C.T. Hill, M. Lindner, \prd{41}{1990}{1647}.

%%%%%%%%% part 1 %%%%%%%%%%%%%%%%%
\bibitem{kaue1}  G.L. Kane and M.E. Peskin, \npb{195}{1982}{29}.
\bibitem{albaj3} C. Albajar et al. [UA1 Coll.],
 Phys. Lett. {\bf B256} (1991) 121, E: ibid. {\bf B262} (1991) 497.
\bibitem{PDG}
{Particle Data Group}, Phys. Rev. {\bf D50},  1171  (1994).


\bibitem{djoua1} A. Djouadi, J.H. K\"uhn and P.M. Zerwas,
               \zpc{46}{1990}{411}.

\bibitem{SuDong} Dong Su, private communication.

\bibitem{Trist} 
TOPAZ Coll., E. Nakano et al., Phys. Lett. {\bf B340} (1994) 135.\\
AMY Coll., F. Liu et al., Phys. Rev. {\bf D49} (1994) 4339.\\
 For a recent review see e.g. K. Miyabayashi, in {\em Recent
electroweak results from TRISTAN} in XXXth Rencontres de Moriond.

\bibitem{SchailZ} D.~Schaile and P.M.~Zerwas, %DESY Report 91--106.
       Phys. Rev. {\bf D45} (1992) 3262.
%%%%%%%%%% mass limits
\bibitem{chano1} M. Chanowitz, M. Furman and I. Hinchliffe,
                 \npb{153}{1979}{402}.
\bibitem{cabib1} N. Cabibbo, L. Maiani, G. Parisi and R. Petronzio,
                 \npb{158}{1979}{296}.
\bibitem{lindn1} M. Lindner, \zpc{31}{1986}{295}.

\bibitem{Munster94} G. M\"unster, 
       Nucl. Phys. B (Proc. Suppl.) {\bf 34} (1994) 596;
ibid., {\bf 42} (1995) 162.

\bibitem{HallThis} L. Hall, these proceedings.

\bibitem{albre1} H. Albrecht et al. (ARGUS), \plb{192}{1987}{245}.
            M. Artuso et al. (CLEO), \prl{62}{1989}{2233}.

\bibitem{veltm1} M. Veltman, \app{8}{1977}{475}.
\bibitem{veltm2} M. Veltman, \npb{123}{1977}{89}.


\bibitem{avdeev} L. Avdeev et al., \plb{336}{1994}{560};
                  Erratum ibid. {\bf B349} (1995) 597.

\bibitem{chetyr} K.G. Chetyrkin, J.H. K\"uhn, and M. Steinhauser,
                  \plb{351}{1995}{331}, \prl{75}{1995}{3394}.


\bibitem{acta} J.H.~K\"uhn, \app{12}{1981}{347}.\\
               J.H.~K\"uhn, \apa{\ Suppl.XXIV}{1982}{203}.


\bibitem{jk3}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \plb{207}{1988}{91}.

\bibitem{smith}
B.~H. Smith and M.~B. Voloshin, \plb{340}{1994}{176}.

\bibitem{ac95a}
A.~Czarnecki, \app{26}{1995}{845}.

\bibitem{BLM}
S.~J. Brodsky, G.~P. Lepage, and P.~B. Mackenzie, \prd{28}{1983}{228}.

\bibitem{den1}
A.~Denner and T.~Sack, \npb{358}{1991}{46}.

\bibitem{eil}
R.~Migneron G.~Eilam, R.R.~Mendel and A.~Soni, \prl{66}{1991}{3105}.

\bibitem{DennerHoang} 
A. Denner and A.H. Hoang, Nucl. Phys. {\bf B397} (1993) 483.

\bibitem{JezK}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \prd{48}{1993}{R1910};
\newblock E: ibid. {\bf D49} (1994) 4970.

\bibitem{Fujii}
K.~Fujii,
\newblock these proceedings.

\bibitem{stre}
{J.H.~K\"uhn} and K.H. Streng, \npb{198}{1982}{71}.

\bibitem{jk1}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \npb{320}{1989}{20}.

\bibitem{cjk}
A.~Czarnecki, M.~Je{\.z}abek, and J.~H. K{\"u}hn, \app{20}{1989}{961},
E: ibid, {\bf B23} (1992) 173.\\
M.~Je{\.z}abek and C. J\"unger, \app{24}{1993}{1923}.

\bibitem{czajk91}
A.~Czarnecki, M.~Je{\.z}abek, and J.~H. K{\"u}hn, \npb{351}{1991}{70}.

\bibitem{cjkk94}
A.~Czarnecki, M.~Je{\.z}abek, J.~G. K{\"o}rner, and J.~H. K{\"u}hn,
\prl{73}{1994}{384}.

\bibitem{cj94}
A.~Czarnecki and M.~Je{\.z}abek, \npb{427}{1994}{3}.

\bibitem{ns30} G.L.~Kane, Proceedings, Madison Workshop 1979; \\
          I.~Bigi et al.\ in Ref.\cite{bigi2}; \\
          J.F.~Gunion, H.E.~Haber, G.~Kane and S.~Dawson, {\it The Higgs
          Hunter's Guide}, Addison--Wesley 1990; \\
          C.S.~Li and T.C.~Yuan, \prl{42}{1990}{3088}.

\bibitem{bigi2}  I. Bigi, Y.L. Dokshitzer, V.A. Khoze,
J. K\"uhn and P. Zerwas, \plb{181}{1986}{157}.

\bibitem{reya1}  E. Reya, P.M. Zerwas [conv.] et al., in Proceedings ECFA
              LHC Workshop, Aachen 1990, CERN 90--10.
\bibitem{PIK}P. Igo-Kemenes, J.H. K\"uhn [conv.], Proceedings,
                {\em \epem\ Collisions at 500~GeV: The  Physics Potential},
                DESY 92-123, pg.~255-326, pg.~327-378.

\bibitem{CzarDav}
A. Czarnecki and S. Davidson,
 Phys. Rev. {\bf D48} (1993) 4183; ibid., {\bf D47} (1993) 3063.\\
A. Czarnecki and A.I. Davydychev,
 \plb{325}{1994}{435}.

\bibitem{EWHiggs}
A. Czarnecki, \prd{48}{1993}{5250}.


\bibitem{ns38} J.~Gunion et al., Ref.\cite{ns30}; \\
          A.~Mendez and A.~Pomarol, \plb{252}{1990}{461}; \\
          C.S.~Li and R.J.~Oakes, \prd{43}{1991}{855}.

\bibitem{ns31} K.~Hikasa and M.~Kobayashi, \prd{36}{1987}{724}; \\
    H.~Baer, M.~Drees, R.~Godbole, J.F.~Gunion and X.~Tata, Tallahassee
    Report FSU--HEP--910308 (1991).
\bibitem{ns39} K.~Hidaka, Y.~Kizukuri and T.~Kon, Gakugei Report TGU--12
          (1991).

% hadron colliders 
\bibitem{ns40} J.~Ellis and S.~Rudaz, \plb{128}{1983}{248}; \\
          A.~Bouquet, J.~Kaplan and C.~Savoy, \npb{262}{1985}{299}.

\bibitem{ns41} G.~Eilam, J.L.~Hewett and A.~Soni, \prd{44}{1991}{1473};

\bibitem{ns32} H.~Fritzsch, \plb{224}{1989}{423}.

\bibitem{ns33} W.~Buchm\"uller and M.~Gronau, \plb{220}{1989}{641}.

\bibitem{gluec1} M. Gl\"uck, J.F. Owens and E. Reya,
                \prd{15}{1978}{2324};\\
                 B.L. Combridge, \npb{151}{1979}{429}.

\bibitem{wille1} S. Willenbrock and D.A. Dicus, \prd{34}{1986}{155};\\
                 C.--P. Yuan, \prd{41}{1990}{42}.

\bibitem{Martin95} A.D. Martin, W.J. Stirling, and R.G. Roberts,
hep-ph/9506423. 

\bibitem{Been91}
         W. Beenakker, W. L. van Neerven, R. Meng, G. A. Schuler and J. Smith,
                   Nucl. Phys. {\bf B351} (1991) 507.
\bibitem{nason1} P. Nason, S. Dawson and R. K. Ellis,
                   \npb{303}{1988}{607};
                 {\em ibid.} {\bf B}327 (1989) 49; Erratum B335 (1990)
                 260.\\
G. Altarelli, M. Diemoz, G. Martinelli, and P. Nason, Nucl. Phys. {\bf
  B308} (1988) 724.

\bibitem{beena1} W. Beenakker, H. Kuijf, W. L. van Neerven and J. Smith,
                    \prd{40}{1989}{54}.
\bibitem{meng1} R. Meng, G. A. Schuler, J. Smith and W. L. van Neerven,
              \npb{339}{1990}{325}.

\bibitem{diemo1} M. Diemoz, F. Ferrari, E. Longo and G. Martinelli,
                    \zpc{39}{1988}{21};\\
           M. Gl\"uck, E. Reya and A. Vogt, \zpc{48}{1990}{471} and
              Proceedings of the 'Small-x' Meeting, DESY, 1990.
\bibitem{Appel} D. Appel, G. Sterman, and P. Mackenzie,
  \npb{309}{1988}{259}. 

\bibitem{Laenen} E. Laenen, J. Smith, and W.L. van Neerven,
  \npb{369}{1992}{543}; \plb{321}{1994}{254}.

\bibitem{Berger} E.L. Berger and H. Contopanagos,
\plb{361}{1995}{115}.

\bibitem{Berends} F.A. Berends, J.B. Tausk, and W.T. Giele,
Phys. Rev. {\bf D47} (1993) 2746.

\bibitem{guesk1} S. G\"usken, J. H. K\"uhn and P. M. Zerwas,
                    \plb{155}{1985}{185}.
\bibitem{fadin1} V. Fadin, V. Khoze, \jl{46}{1987}{417};
                   \sovnp{48}{1988}{487}; \\
                   V. Fadin, V. Khoze and T. Sj\"ostrand,
 \zpc{48}{1990}{467}.
\bibitem{holli2} W. Hollik et al.~in Ref.\cite{reya1};\\
              W. Beenakker, A. Denner, W. Hollik et al., R. Mertig, T. Sack,
 and D. Wackeroth, \npb{411}{1994}{343}.

\bibitem{Lampe} B. Lampe, \plb{348}{1995}{196}.

\bibitem{Barger} V. Barger et al., \prd{35}{1987}{3366}.

\bibitem{KM} J.H. K\"uhn and E. Mirkes, \plb{311}{1993}{458}.

\bibitem{KP} J.H. K\"uhn and M. Peter, \zpc{67}{1995}{271}.

\bibitem{Letter} J.H. K\"uhn and E. Mirkes, \plb{296}{1992}{425}.
\bibitem{Paper} J.H. K\"uhn and E. Mirkes, \prd{48}{1993}{179}.


\bibitem{lighthiggs} A. Djouadi, M. Spira, and P.M. Zerwas,
  \plb{264}{1991}{440}. 

% part3 

\bibitem{fusi} O.J.P.~Eboli, G.C.~Marques, S.F.~Novales and
                A.A.~Natale, \prd{34}{1986}{771},\\
               S.~Dawson and S.~Willenbrock, \npb{284}{1987}{449},\\
               C.P.~Yuan, \npb{310}{1988}{1},\\
               R.P.~Kauffman, \prd{41}{1990}{3343},\\
                H.~Veltman, \prd{43}{1991}{2236}.
\bibitem{hagi}
       K.~Hagiwara, H.~Murayama and T.~Watanabe, \npb{367}{1991}{257}.
\bibitem{djo} A.~Djouadi, J.~Kalinowski and P.M.~Zerwas,
                \zpc{54}{1992}{255}.

\bibitem{DESY} J.H. K\"uhn, E. Mirkes, and J. Steegborn,
\zpc{57}{1993}{615}.. 

\bibitem{con} M.~Consoli, W.~Hollik and F.~Jegerlehner, in ``$Z$ Physics
              at LEP~1'', ed. G.~Altarelli, CERN89-08.
\bibitem{exac}
          J.~Jersak, E.~Laermann and P.M.~Zerwas,
          \prd{25}{1982}{363},\\
J.~Schwinger, ``Particles, Sources and Fields'', Addison-Wesley (1973),\\
         L.~Reinders, H.~Rubinstein and S.~Yazaki, \prc{127}{1985}{1}.

\bibitem{KueZer} J.H. K\"uhn and P.M. Zerwas, in {\em Advanced Series
    on Directions in HEP}, Vol.~10,  {\em Heavy Flavours},
  Eds. A.J. Buras and M. Lindner, World Scientific, 1992, p.~434.

\bibitem{appr} J.~Schwinger in \cite{exac}, and ref.~\cite{guesk1}.

\bibitem{HKT} A.H. Hoang, J.H. K\"uhn, and T. Teubner,
\npb{452}{1995}{173}. 

\bibitem{CKS} K. Chetyrkin, J.H. K\"uhn, and M. Steinhauser,
\plb{371}{1996}{93}. 

\bibitem{jez} M.~Je\.zabek, \zpc{56}{1992}{285}.
\bibitem{jad} S.~Jadach and B.F.L.~Ward, \cpc{56}{1990}{351}.
\bibitem{been1} W.~Beenakker, S.C.~van~der~Marck
              and W.~Hollik, \npb{365}{1991}{24}.

\bibitem{JKsim}
M. Je\.zabek and J.H. K\"uhn, Phys. Lett. {\bf B316} (1993) 360.
\bibitem{BH} W. Beenakker and W. Hollik, \plb{269}{1991}{425}.

\bibitem{GKKS}
B. Grzadkowski, P. Krawczyk, J.H. K\"uhn, and R.G. Stuart, Nucl. Phys. 
{\bf B281} (1987) 18.
\bibitem{fr1} C.~Peterson, D.~Schlatter, I.~Schmitt and P.M.~Zerwas,
               \pr{27}{83}{105}.
\bibitem{fr3} T.~Sj\"ostrand and P.M.~Zerwas, in Proceedings, {\em $e^+e^-$
       Collisions at 500 GeV: The Physics Potential}, DESY 92-123,
       pg.~463-475. 

\bibitem{fr2} G.~Jikia, \plb{257}{1991}{196}.
\bibitem{KOS} V.A. Khoze, L.H. Orr, and J. Sterling,
\npb{378}{1992}{413}. 

\bibitem{st1} G.L.~Kane, G.A.~Ladinsky and C.--P.~Yuan,
              \prd{45}{1992}{124}.\\
              C.--P.~Yuan, \prd{45}{1992}{782}.
\bibitem{BerOve} W. Bernreuther and P. Overmann, Proc. of the Workshop
  $e^+e^-$ Collisions at 500 GeV, the Physics Potential, Part C,
  P.M. Zerwas (ed.), DESY 93-123C, p. 287, and refs.~therein.

\bibitem{st2} C.~Schmidt and M.E.~Peskin, \prl{69}{1992}{410}.


\bibitem{kuehn2} J.H. K\"uhn, A. Reiter and P.M. Zerwas,
                   \npb{272}{1986}{560}.

\bibitem{st4} A.~Devoto, J.~Pumplin, W.~Repko and G.L.~Kane,
            \prl{43}{79}{1062}; 

\bibitem{jk41} J.H. K\"uhn, \npb{237}{1984}{77}.

%\bibitem{JKunp}
%      M.~Je\.zabek and J.H.~K\"uhn, Karlsruhe Univ. preprint TTP92-10.

\bibitem{Landau}
L. Landau and E. Lifshitz, {\em Non-relativistic Quantum Mechanics}, sect.~36.
\bibitem{Messiah}
A. Messiah, {\em Quantum Mechanics}, Chapter XI, sect.~7-11 
and Appendix B, sect.~4.

\bibitem{fk}
V.S.~Fadin and V.A.~Khoze, \jet{46}{1987}{525},


\bibitem{apple}
T.~Appelquist, M.~Dine and I.~Muzinich, \prd{17}{1978}{2074}.
\bibitem{sus}
L.~Susskind, in ``Weak and Electromagnetic Interactions at High
Energies'', Les Houches (1977). 
\bibitem{fis}
W.~Fischler, \npb{129}{1977}{157}.
\bibitem{bil}
A.~Billoire, \plb{92}{1980}{343}.
\bibitem{bern}
W.~Bernreuther and W.~Wetzel, \npb{197}{1982}{228}.
\bibitem{marc}
W.J.~Marciano, \prd{29}{1984}{580}.
\bibitem{JKT}  M.~Je\.zabek, J.H.~K\"uhn and T.~Teubner,
   \zpc{56}{1992}{653}

\bibitem{pesk}
J.M.~Strassler and M.E.~Peskin, \prd{43}{1991}{1500}.
\bibitem{TJ} M.~Je\.zabek and T.~Teubner, \zpc{59}{1993}{669}
\bibitem{Buchm}
For a pedagogical discussion see e.g.~the 
appendix of W. Buchm\"uller and S.-H. H. Tye, Phys. Rev. {\bf D24} (1981) 132.
\bibitem{Igo}
P. Igo-Kemenes, M. Martinez, R. Miguel, and S. Orteu,  in Proceedings,
{\em \epem\ Collisions 
at 500 GeV: The Physics Potential}, Part C; ed. P.M. Zerwas, 
DESY 93-123C, pg. 319.
\bibitem{GuthK}
R. Guth and J.H. K\"uhn,             Nucl. Phys. 
{\bf B368} (1992) 38.

\bibitem{JKm}
M. Je\.zabek and J.H. K\"uhn,  Phys. Lett. {\bf B207} (1988) 91.


\bibitem{Sumino} Y.~Sumino, K.~Fujii, K.~Hagiwara, H.~Murayama, C.-K.~Ng,
   \prd{47}{1992}{56}

\bibitem{JKTZer} M.~Je\.zabek, J.H.~K\"uhn and T.~Teubner,
 in Proceedings, {\em $e^+e^-$
       Collisions at 500 GeV: The Physics Potential}, DESY 92-123C,
pg.~303. 

\bibitem{peskin}  J.M.~Strassler and M.E.~Peskin, \prd{43}{1991}{1500}
\bibitem{MurSum2}
    H. Murayama and Y. Sumino, \prd{47}{1993}{82}.

\bibitem{hjkt}
    R. Harlander, M. Je\.zabek, J.H. K\"uhn and T. Teubner,
    \plb{346}{1995}{137}.

\bibitem{krz}
    J.H. K\"uhn, A. Reiter and P.M. Zerwas, \npb{272}{1986}{560}.
\bibitem{dev}
    A. Devoto, J. Pumplin, W. Repko and G.L. Kane,
    \prl{43}{1979}{1062}.
\bibitem{FKotsky}
    V.S. Fadin, V.A. Khoze and M.I. Kotsky, \zpc{64}{1994}{45}.

\bibitem{ttp48}
R. Harlander, M. Je\.zabek, J.H. K\"uhn, and M. Peter, Karlsruhe
preprint TTP 95-48, hep-ph/9604328, in print in Z.~Phys. {\bf C}.
\bibitem{MelYak1} K. Melnikov and O. Yakovlev, \plb{324}{1994}{217}.
\bibitem{sumph} Y. Sumino, PhD thesis, University of Tokyo, 1993
(unpublished). 
\bibitem{Huff} R.W.  Huff, {\it Ann. Phys.} {\bf 16} (1961) 288.
\bibitem{FMY94}
    K. Fujii, T. Matsui and Y. Sumino, \prd{50}{1994}{4341}.

\bibitem{Uber} H. \"Uberall, {\it Phys. Rev.} {\bf 119} (1960) 365. 
\bibitem{MKum} W. M\"odritsch and W. Kummer, \npb{430}{1994}{3}.


\end{thebibliography}

\chapter{Top quarks in $e^+e^-$ annihilation}

A variety of reactions is conceivable for top quark production
at an electron positron collider.  Characteristic Feynman
diagrams are shown in
Fig.\ref{fig:3.1}.
% the resulting cross sections in Fig.\ref{fig:3.2}.
$e^+ e^-$ annihilation through the virtual photon and
$Z$ (Fig.\ref{fig:3.1}a)
dominates and constitutes the reaction of interest for the currently
envisaged energy region.
\begin{figure}
\vspace{10cm}
\caption{Feynman diagrams for
 $t\bar{t}$ or $t\bar{b}$ production.}

\label{fig:3.1}
\end{figure}

In addition one may also consider \cite{fusi} a variety of
gauge boson fusion reactions (Fig.\ref{fig:3.1}b-d) that
are in close analogy to $\gamma \gamma$
fusion into hadrons at $e^+e^-$ machines of lower energy.
Specifically these are single top production,
\begin{equation}
e^+e^- \to \bar{\nu} e^- t \bar{b}
\label{eq:26}
\end{equation}
or its charge conjugate and top pair production through neutral or charged
gauge boson fusion
\begin{eqnarray}
e^+e^- & \to & e^+e^- t \bar{t} \nonumber\\
e^+e^- & \to & \bar{\nu} \nu t \bar{t}
\label{eq:27}
\end{eqnarray}

The experimental observation of these reactions would allow
to determine the coupling of top quarks to gauge bosons,
in particular also to longitudinal $W$ bosons and $Z$ bosons,
in the space-like region and eventually at large momentum transfers.
This would constitute a nontrivial test of the mechanism
of spontaneous symmetry breaking.

The various cross sections increase with energy in close analogy
to $\gamma \gamma$ reactions, and eventually even exceed $e^+e^-$
annihilation rates.  However, at energies accessible in the
foreseeable future these reactions are completely
negligible:  for an integrated luminosity of $10^{40}$ cm$^{-2}$,
at $E_{cm} = 500~GeV$ and for $m_t = 150~GeV$ one expects about
one $e^+e^ - t\bar{t}$ event (still dominated by
$\gamma \gamma$ fusion). At that same energy the cross sections for
$e^+ \nu \bar{t} b + c.c.$ and $\nu \bar{\nu} t \bar{t}$ final states
are still one to two orders of magnitude smaller.

Another interesting class of reactions is $e^+e^-$ annihilation
into heavy quarks in association with gauge or Higgs bosons:
\begin{eqnarray}
e^+ e^- & \to & t \bar{t} Z \\
%\label{eq:28}
e^+ e^- & \to & t \bar{b} W^- \\
%\label{eq:29}
e^+ e^- & \to & t \bar{t} H \\
%\label{eq:30}
e^+ e^- & \to & t \bar{b} H^-
\label{eq:31}
\end{eqnarray}

Two amplitudes contribute to the first reaction \cite{hagi}:
The $t \bar{t}$ system may be produced through a virtual Higgs
boson which by itself was radiated from a $Z$ (Fig.\ref{fig:3.2}).
The corresponding amplitude dominates the rate and
provides a direct measurement of the Yukawa coupling.
The radiation of longitudinal $Z$'s from the quark line
in principle also carries information on the symmetry
breaking mechanism of the theory.
\begin{figure}
\vspace{50mm}
\caption{\label{fig:3.2}
Amplitudes relevant for
$e^+e^- \to t\bar{t}Z$ and for $e^+e^- \to t\bar{t}H$.}
\end{figure}
The transverse part of the $t \bar{t} Z$ coupling,
i.e. the gauge part, can
be measured directly through the cross section or
various asymmetries in
$e^+e^- \to t \bar{t}$.  The longitudinal part, however,
could only be isolated
with $t \bar{t} Z$ final states.  For an integrated
luminosity of $10^{40}$ cm$^{-2}$
one expects only about 40 events (see sect. 3.1.7)
and it is therefore not
clear whether these can be filtered from the huge background
and eventually used for a detailed analysis.

Light Higgs bosons may be produced in conjunction with
$t \bar{t}$ \cite{djo}. They are radiated either from
the virtual $Z$ with an amplitude that is present
also for massless fermions or directly from heavy quarks as a consequence of
the large Yukawa coupling (Fig.\ref{fig:3.2}).
The latter dominates by far and may therefore be tested
specifically with heavy quark final states.  The predictions for the
rate will be discussed in sect.~3.1.7.  Depending on the mass of the
Higgs and the top 
quark, the reaction could perhaps be detected with an integrated luminosity
of $10^{40}cm^{-2}$.

Top quark production in $\gamma \gamma$ collisions is conceivable at a
``Compton collider''.  It requires special experimental provisions for the
conversion of electron beams into well-focused beams of energetic photons
through rescattering of laser light.  A detailed discussion can be
found in \cite{DESY}. 

Chapter 3 will be entirely devoted to $t\bar t $ 
production in $e^+e^-$ annihilation.  Section 3.1 will be 
concerned with the energy region far above threshold --- with electroweak 
aspects as well as with specific aspects of top hadronisation. The emphasis 
of section 3.2 will be on the threshold region which is governed by the 
interplay between bound state formation and the rapid top decay.



\section{Top production above threshold}
\subsection{Born predictions}
From the preceding discussion it is evident that the bulk of top studies
at an $e^+ e^-$ collider will rely on quarks produced in $e^+ e^-$ annihilation
through the virtual $\gamma$ and Z, with a production cross section of the
order of $\sigma _{point}$.  For quarks tagged at an angle $\vartheta $, the
differential cross section in Born approximation is a binomial in $\cos \vartheta $
\begin{equation}
\frac{d\sigma}{d \cos\vartheta} = \frac{3}{8} \left( 1 + \cos^2 \vartheta \right) \sigma _U + \frac{3}{4} \sin^2 \vartheta \sigma _L + \frac{3}{4} \cos \vartheta \sigma _F
\label{eq:32}
\end{equation}
$U$ and $L$ denote the contributions of unpolarized and longitudinally polarized
gauge bosons along the $\vartheta$ axis, and $F$ denotes the difference
between right and left polarizations.  The total cross section is the sum
of $U$ and $L$,
\begin{equation}
\sigma = \sigma _U + \sigma _L
\label{eq:33}
\end{equation}
the forward/backward asymmetry is given by the ratio
\begin{equation}
A^{FB} = \frac{3}{4} \frac{\sigma _F}{\sigma}
\label{eq:34}
\end{equation}

The $\sigma^i$ can be expressed in terms of the cross sections for the
massless case in Born approximation,
\begin{eqnarray}
\sigma ^U_B & = & \beta \sigma^{VV} + \beta ^3 \sigma ^{AA} \nonumber \\
\sigma ^L_B & = & \frac{1}{2} \left( 1 - \beta ^2 \right) \beta \sigma^{VV} \nonumber \\
\sigma ^F_B & = & \beta ^2 \sigma^{VA}
\label{eq:35}
\end{eqnarray}
with
\begin{eqnarray}
\sigma ^{VV} & = & \frac{4 \pi \alpha^2 (s) e^2_e e^2_Q}{s} \nonumber \\
& & + \frac{G_F (s)}{\sqrt{2}} e_e e_Q (\upsilon _e + \rho a_e )
\upsilon _Q \frac{m^2_Z \left( s-m^2_Z \right)}
{\left( s-m^2_Z \right)^2 + \left(\frac{s}{m_Z}
\Gamma _Z \right)^2} \nonumber \\
& & + \frac{G^2_F}{32 \pi} \left( \upsilon ^2_e
+ a^2_e + 2 \rho \upsilon _e a_e \right) \upsilon ^2_Q \frac{m^4_Z s}
{\left( s-m^2_Z \right)^2
+ \left(\frac{s}{m_Z} \Gamma _Z \right)^2} \nonumber \\
\sigma ^{AA} & = & \frac{G^2_F}{32 \pi}
\left( \upsilon ^2_e + a^2_e + 2 \rho \upsilon _e a_e \right)
a^2_Q \frac{m^4_Z s}{\left( s-m^2_Z \right)^2
+ \left(\frac{s}{m_Z} \Gamma _Z \right)^2} \nonumber \\
\sigma ^{VA} & = & \frac{G_F \alpha (s)}{\sqrt{2}} e_e
(\rho \upsilon _e + a_e )e_Q a_Q \frac{m^2_Z
\left( s-m^2_Z \right)}{\left( s-m^2_Z \right)^2
+ \left(\frac{s}{m_Z} \Gamma _Z \right)^2} \nonumber\\
& & + \frac{G^2_F}{16 \pi} \left( 2 \upsilon _e a_e
+ \rho (\upsilon ^2_e + a^2_e)\right) \upsilon_Q a_Q
\frac{m^4_Z s}{\left( s-m^2_Z\right)^2
+ \left(\frac{s}{m_Z}\Gamma _Z \right)^2}
\label{eq:36}
\end{eqnarray}

The fermion couplings are given by
\begin{eqnarray}
\upsilon_f = 2I^f_3 - 4e_f \sin ^2 \theta _w \quad , \qquad a_f = 2I^f_3
\label{eq:37}
\end{eqnarray}
and the possibility of longitudinal electron polarization
($\rho = -1;+1;0$ for
right\-handed; lefthanded; unpolarized electrons) has been included.
Alternatively one may replace $G_F m^2_Z$ by
\begin{equation}
G_F m^2_Z = \frac{\pi \alpha (s)}{\sqrt{2} \sin^2 \theta_W \cos^2 \theta _W}
\label{eq:38}
\end{equation}
With $\sin^2\theta_W$ ($\approx$0.23)
interpreted as $\sin^2 \theta_{eff}$ \cite{con}
this formula accommodates the leading logarithms from the running coupling
constant as well as the quadratic top mass terms in the threshold region.  

\subsection{Radiative corrections}

\underline{QCD corrections}
to this formula are available for arbitrary $m^2 / s$ up
to first order in $\alpha _s$:
\begin{eqnarray}
\sigma & = & \frac{(3 - \beta ^2)}{2} \beta \sigma^{VV} \left( 1 + \frac{4}{3} \frac{\alpha _s}{\pi} K_V \right) \nonumber \\
& & + \beta ^3 \sigma ^{AA} \left( 1 + \frac{4}{3} \frac{\alpha _s}{\pi} K_A \right)
\label{eq:40}
\end{eqnarray}
The exact result \cite{exac} for $K_{V,A}$ can be found in \cite{KueZer}.
These  QCD enhancement factors are
well approximated by \cite{appr}
\begin{eqnarray}
K_V & = & 1 + \frac{4}{3} \alpha_s
\left[ \frac{\pi}{2\beta} - \frac{3+\beta}{4}
\left( \frac{\pi}{2} - \frac{3}{4\pi} \right) \right] \nonumber \\
K_A & = & 1 + \frac{4}{3} \alpha_s
\left[ \frac{\pi}{2\beta} -
\left(\frac{19}{10} - \frac{22}{5}\beta +\frac{7}{2}\beta^2\right)
\left( \frac{\pi}{2} - \frac{3}{4\pi} \right) \right] \nonumber \\
\alpha_s & = & \frac{12\pi}{25 \log (4p^2_t / \Lambda ^2)}
\label{eq:43}
\end{eqnarray}
The next to leading order corrections 
to $K_V$ were calculated only recently \cite{HKT,CKS}.
The scale in $\alpha _s$ chosen
above was guessed on the basis of general
arguments \cite{guesk1} which were confirmed by the forementioned
complete calculations.
\par\noindent
For small $\beta$ these factors develop the familiar Couloumb
enhancement $\sim{2\pi\alpha_s\over 3\beta}$, compensating the
phase space $\sim \beta$.  This leads to a nonvanishing cross
section which smoothly joins the resonance region.  Details of
this transition will be treated in section 3.2.

To prepare this discussion, let us briefly study the limit of applicability 
of fixed order perturbation theory.  The leading terms in the perturbative 
expansion close to threshold are obtained from Sommerfeld's rescattering 
formula $\left(x\equiv {4\over 3}{\pi\alpha\over \beta}\right)$
\be
K_V^{\rm Som} = {x\over 1 - e^{-x}} &=& 1 +{x\over 2} + {B_1 x^2\over 2!} 
- {B_2 x^4\over 4!}+  {B_3 x^6\over 6!}\pm\ldots
\nonumber \\
&=& 1+{x\over 2} + { x^2\over 12} - {x^4\over 720} + {x^6\over 
5040}\pm\ldots
\ee
with $B_i$ being the Bernoulli numbers.  At first glance one might require 
$x\le 1$ for the perturbative expansion to be valid.  However, 
significantly larger values of $x$ are acceptable. The full  Sommerfeld 
factor $K_V^{\rm Som}$ is remarkably well approximated by the first three 
terms of the series for surprisingly large $x$ (only 6\% deviation for 
$x=4!$).  For top quarks this corresponds to $\beta\approx 0.13-0.14$ 
and hence down to about 3 GeV above the nominal threshold.  Upon closer 
inspection one also observes that the formula given in
eq.~(\ref{eq:43}) (a result  
of order $\alpha_s$) coincides numerically well with the correction factor 
$K_V^{\rm Som} \left(1-{16\over 3}{\alpha_s\over \pi}\right)$ which 
incorporates rescattering and hard gluon vertex corrections.  The results 
presented in these lectures are based on the Born predictions plus ${\cal 
O} (\alpha_s)$ corrections.


\underline{Initial state radiation}
has an important influence on the magnitude
of the cross section.  $\sigma(s_{eff})$ is folded with
the Bonneau Martin structure function, supplemented by the
summation of large logarithms.  A convenient formula for
the non-singlet structure function in the leading logarithmic
approximation has been obtained in \cite{jez},
which is a natural extension of a formula
proposed in \cite{jad}.
%
This leads to a significant suppression by about a factor
\begin{equation}
\left( \frac{\delta W}{m_t} \right) ^{ {2\alpha\over \pi}\left(
\ln{m_t^2\over m_e^2}-1\right)}
 \approx 0.5-0.6
\label{eq:47}
\end{equation}
with $\delta W = 1-5~GeV$ in the resonance
and threshold region. The correction factor increases rapidly
with energy, but stays below 0.9 in
the full range under consideration (Fig.\ref{fig:3.5}).



\underline{Electroweak corrections}
to the production cross section
in the continuum have been studied in \cite{been1}.  Apart from
a small region close to threshold they are negative.
Relative to the $G_F$ parametrized Born approximation they
decrease the cross section by -6.3\% to -9.3\%, if $m_t$ is
varied between 100 and 200~GeV, $m_H$ between 42 and 1000~GeV,
and $E_{cm}$ fixed at 500~GeV.  QCD and electroweak
corrections are thus of equal importance (Fig.~\ref{fig:3.6}).

Close to threshold and for relatively small Higgs boson masses a rapid 
increase of these corretions is observed (Fig.~\ref{fig:3.7}) which
can be attributed  
to the attractive Yukawa potential induced by light Higgs boson exchange.  
Several GeV above threshold, and for $m_H$ around or below 100 GeV it is 
more appropriate to split these corrections into hard and soft exchange and 
incorporate the latter in an instantaneous Yukawa potential \cite{JKsim}.

\begin{figure}[ht]
  \begin{center}
    \leavevmode
    \epsfxsize=12.cm
    \epsffile[65 220 410 450]{fig_3.3.ps}
    \hfill
\caption{Cross section for $t\bar{t}$ production,
including resonances, QCD corrections
and initial state radiation in units of $\sigma_{point}$.}
\label{fig:3.5}
  \end{center}
\end{figure}
\begin{figure}
\vspace{80mm}
\caption[]{Genuine electroweak
corrections to top production in $e^+e^-$ annihilation.
From \cite{been1}.}
\label{fig:3.6}
\end{figure}
\begin{figure}
\vspace{80mm}
\caption{Relative size (in percent) of electroweak corrections in the
threshold region for $m_t=200$ GeV and different Higgs masses (from
\protect\cite{BH}).}
\label{fig:3.7}
\end{figure}

\subsubsection{Longitudinal polarization}

It should be mentioned that linear colliders might well operate to
a large extent with polarized (electron) beams.  The cross section for
this case can be derived from (\ref{eq:36}).  For top quarks the resulting
right/left asymmetry
\begin{equation}
A_{LR} =  (\sigma _L - \sigma _R) / (\sigma _L + \sigma _R )
\label{eq:48}
\end{equation}
is sizable (Fig.\ref{fig:RL}) and amounts to about $-\,0.4$,
\begin{figure}[ht]
  \begin{center}
    \leavevmode
    \epsfxsize=12.cm
    \epsffile[85 280 510 565]{fig_3.6.ps}
    \hfill
\caption{Right/left asymmetry as function
of $E_{cm}$ for $m_t = 150~GeV$.}
\label{fig:RL}
  \end{center}
\end{figure}
reducing the production cross section with righthanded electrons.
However, selection of righthanded electron beams decreases the $W$ pair
cross section even stronger, thereby enhancing the top quark
signal even before cuts are applied.
Electroweak corrections to $A_{LR}$ in the threshold region have been
calculated in \cite{GKKS}.

\subsection{Top quark fragmentation}

The experimental analysis of charm and bottom fragmentation functions has
clearly demonstrated that heavy quark fragmentation is hard in
contrast to the fragmentation of light
quarks. This is a consequence of the inertia of heavy particles, the momentum
of which is not altered much if a light quark $\bar q$ is attached to
the heavy quark $Q$ in the fragmentation process
to form a bound state $(Q\bar q)$, see e.g.\ \cite{fr1}. At the same
time soft infrared gluon radiation is damped if the color source is
heavy.

For $m_t \ge 100$ GeV the strong fragmentation process and
the weak decay mechanism are intimately intertwined \cite{fr3}. The
lifetime 
$\tau_* < \Lambda^{-1}$ becomes so short that the mesonic $(t\bar q)$
and baryonic $(tqq)$ bound states cannot be built--up anymore.
Depending on the initial top quark energy, even remnants of the $t$
quark jet may not form anymore \cite{bigi2}. Hadrons can be created in
the string stretched between the $t$ and the $\bar t$ only if the
quarks are separated by about 1 fermi before they decay. If the flight
path $\gamma \tau_*$ is less than $1/2$ fm,
the length of  the $t-\bar t$ string is too short to form
hadrons and jets cannot develop anymore along the flight direction
of the top quarks.  For $m_t\approx 180$ GeV  top quark energies above
1 TeV are required to allow nonperturbative strings between $t$ and
$\bar t$.  
``Early'' nonperturbative production of particles from the string between 
$t$ and $\bar t$ is thus absent for all realistic experimental 
configurations.  ``Late'' production from the $b$ and $\bar b $ jets 
produced in top decays dominates.  
\begin{figure}
\vspace{155mm}
\caption{Rapidity distributions of ``early'' particles (full) and
``late'' ones (dashed) for three different top masses: 90 GeV in a)
100 GeV in b) and 120 GeV in c).  (From \protect\cite{fr3}.)}
\label{fig:SjZ}
\end{figure}
This is illustrated in Fig.~\ref{fig:SjZ}.  Early 
production dominates for $m_t = 90$ GeV, late production for $m_t=120$ GeV 
and {\it a forteriori} for the actual value around 180 GeV. 


The perturbative radiation of soft gluons, too, is interrupted by the
$t$ quark decay \cite{fr2}.
The angular distribution $(\Theta)$ and the
energy distribution $(\omega)$ of the radiated gluons is
approximately given by
\begin{equation}
dP_g = \frac{4\alpha_s}{3\pi} \frac{\Theta^2 d\Theta^2}{\left[
\Theta^2 + \frac{1}{\gamma^2} \right]^2 + \left[ \frac{\Gamma}{\gamma
\omega} \right]^2 } \frac{d\omega}{\omega}
\end{equation}
\noindent for a short--lived radiation source accelerated to $\gamma =
E_t/m_t$. The gluons accumulate on the surface of a cone with
half--aperture $\Theta_c \sim \gamma^{-1}$ for a long--lived $t$, but
$\sim \gamma^{-1} \sqrt{\gamma\Gamma/\omega}$ if the particle decays
quickly. The energy spectrum rises from zero to a maximum at $\omega
\sim \gamma\Gamma$ before falling off $\sim 1/\omega$ for large $\omega$,
if the width is greater than the confinement scale $\Lambda$.

The impact of the finite width on the angular distribution of gluon 
radiation will be visible if $\omega \le \Gamma_t E_t/m_t$. For a linear 
collider with c.m.~energy of 2 TeV gluon jets with energies of 10 GeV and 
below would be affected \cite{KOS}.  The radiation pattern  is shown in
\begin{figure}
\vspace{10cm}
\caption{Sensitivity of the soft gluon distribution to the top width
for back-to-backk $t$ and $b$: $dN/d\cos(\theta_g)$ for $M$ = 140 GeV,
$\theta_b=180^\circ$, and $\Gamma$ = 0, 0.7, 5, 20 GeV, and $\infty$
as marked. $W$ = 1000 GeV. (From \protect\cite{KOS}.)}
\label{fig:KhozeOrrSt}
\end{figure}
Fig.~\ref{fig:KhozeOrrSt}  for $m_t = 140$  GeV and  
$\sqrt{s} = 600 $ GeV, with $\Gamma_t$ tuned to different values in order 
to demonstrate the sensitivity of such a measurement.  The impact of the 
conversion of gluons to hadrons has been ignored in this study.  The 
picture is, further, complicated by the interference between radiation from 
top production and decay --- a phenomen characteristic for unstable 
particles. These phenomena allow to probe the time evolution of 
hadronisation in a unique manner.  Their understanding is a necessary 
prerequisite for any top mass measurement through a kinematic analysis of 
$b$-jet-$W$ final states.

%%%%%%%%%%%%%Nur in Review%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Static $t$ parameters}
%
Because of the large $t$ mass, deviations from the Standard
Model may manifest themselves in the top quark sector first.
Examples in which the large mass is crucial are provided by
multi--Higgs doublet models, models of dynamical symmetry breaking and
compositeness. These effects can globally be described by form factors
parametrizing the electroweak $t\bar t$ production current $(a =
\gamma,Z)$ and the weak $(t,b)$ decay
current $(a=-)$ \cite{st1,BerOve},
\begin{equation}
j_\mu^a \sim F_{1L}^a \gamma_\mu P_L + F_{1R}^a \gamma_\mu P_R +
\frac{i\sigma_{\mu\nu} q_\nu}{2m_t} [ F_{2L}^a P_L + F_{2R}^a P_R]
\end{equation}
\noindent [$P_{L,R}$ project on the
left and right chirality components of the wave
functions.] In the \sm, $F_{1L}^- = 1$ while all other $F^-_i$ vanish;
$F_{1L}^\gamma = F_{1R}^\gamma = 1$ and $F_{2L}^\gamma = F_{2R}^\gamma
= 0$, analogously for the $Z$ current. ${\cal{CP}}$ invariance
requires $F_{2L}^{\gamma,Z} = F_{2R}^{\gamma,Z}$ in the $t\bar t$
production current, and equal phases for $F_{1L}^-$ and $F_{2R}^-$
{\it etc.}\ in the decay current. The static values of the form factors
$F_2^{\gamma,Z}$ are the anomalous magnetic and electric dipole moments
of the top quark.

The form factors are determined experimentally by measuring the
angular distribution of the $t\bar t$ decay products, $e^+e^- \ra
t\bar t, t \ra b W^+, W^+ \ra f \bar f' $ {\it etc.}
This requires the top quark to be treated as a free particle,
the polarization of which not being affected by non--perturbative
hadronic binding effects. This assumption is justified by the short
lifetime of the top quark as discussed earlier. Details of the general
helicity analysis can be found in the literature \cite{st1,st2}.


\subsection{Normal polarization of the top quarks}
%
A non--zero component of the $t$ polarization vector that is
normal to the production plane can be generated only by the
interference between complex helicity flip and non--flip
amplitudes. Such relative phases can arise from ${\cal{CP}}$ violation
but also from higher order loop corrections due to gluon exchange in
the final state \cite{kuehn2,st4,st1}
or electroweak corrections involving
Higgs and gauge bosons \cite{st1}. The QCD induced normal polarization
is generally less than 5\%, the electroweak normal polarization is
smaller still.
[By contrast, longitudinal and transverse polarization components within the
$t\bar{t}$ production plane are generated already at the tree level of
the electroweak interactions and they are large in general; see
\cite{kuehn2} for the discussion of details.]

\subsection{Angular correlations of $t\bar t$ decay products}
As stated in the previous chapter, top quarks are produced through the
virtual photon and $Z$. In the threshold region they are polarized to a
degree
\begin{equation}
P_t= A_{RL}\approx - 0.4
\end{equation}
Assuming for the distribution of leptons from the decay of polarized
top quarks
\begin{equation}
\frac{dN}{dx\,d\cos\theta}=f(x)+ g(x) P_t\cos\theta
\end{equation}
(with $g(x)=f(x)$ in the \sm, see eq.\ref{eq:15})
the angular distribution 
allows to test for the chirality of the $tb$
current. Implicitly it was assumed that
hadronization does not affect
the top spin degrees of freedom \cite{acta,jk41}. This assumption can
be tested independently through the study of correlations between $t$ and
$\bar t$ decay products. In the threshold region the spins are
correlated
$\propto (1+\frac{1}{3} \vec{s_+}\cdot\vec{s_-})$. This leads to the
following correlated $\ell_+\ell_-$ distribution:
\begin{equation}
\frac{dN}{dx_+dx_-d\cos\theta_{+-}}= f_+(x_+)f_-(x_-) +
                      \frac{1}{3} g_+(x_+)g_-(x_-)\cos\theta_{+-}
\end{equation}
where $f_+=f_-$ and $g_+= -g_-$. $\theta_{+-}$ denotes the angle between
$\ell_+$ and $\ell_-$. After averaging the lepton energies,
\begin{equation}
\frac{dN}{d\cos\theta_{+-}}=1+\frac{1}{3} h_+ h_- \cos\theta_{+-}
\end{equation}
Note that the coefficient of the correlation term is $-h^2_+/3$ and
hence always negative (assuming CP conservation). Since
$|h_+|\leq1$ it ranges between 0 and $-1/3$. This limiting value is
assumed in the \sm. A detailed discussion with illustrative examples is
given in \cite{PIK}.


\subsection{Testing the Yukawa Coupling}

With its relatively large Yukawa coupling $g_Y=\sqrt{2} m_t/v\approx
1$  
the top quark 
is uniquely suited to test one of the basic ingredients of the Standard 
Model, the coupling between top quarks and the Higgs boson.  The 
verification of this crucial prediction would confirm the mechanism for the 
generation of fermion masses and hence complete the measurement and 
analysis of basic couplings.  Alternatively, any deviation would provide 
unambiguous proof for new physics.

Different strategies are at hand at an \epem collider which are closely 
tied to the cms energy available and to the mass of the Higgs boson.  For 
relatively light bosons a variety of possibilities appear to be promising: 
vertex corrections affect the cross section for $t\bar t$ production in the 
threshold region.  For a collider in its early stage with an energy around 
500 GeV it may well be the only option available and will be discussed more 
thoroughly in section 3.2.  For higher energies, say around  1 TeV, a 
promising choice is the radiation of Higgs 
bosons from $t\bar t$ (see Fig.~\ref{fig:3.2}) 
with a cross section around 1 fb 
\begin{figure}[htb]
\psfig{figure=djouadi.ps,angle=-90,width=60mm,bbllx=60pt,bblly=60pt,%
bburx=510pt,bbury=340pt}
\caption{The cross section $\sigma(e^+e^- \to t\bar t H)$ (from
\protect\cite{djo}).} 
\label{fig:sigma}
\end{figure}
(Fig.~\ref{fig:sigma}) \cite{djo}. Alternatively one 
may analyse the top quark final states in conjunction with a $Z$ boson.  
This reaction receives important contributions from the left one of
the diagrams depicted in 
Fig.~\ref{fig:3.2}
if the Higgs mass happens to be relatively close to 2$m_t$
\cite{hagi}.
  Again, for 
simple kinematical reasons, high energies are crucial for the reaction to 
be accessible (Fig.~\ref{fig:Hagi}). 
\begin{figure}
\vspace{15cm}
\caption{The Higgs mass ($m_H$) dependences of the total cross
sections of $e^+e^- \to t\bar t Z$ for various top quark masses $m_t$.
The c.m.~energy $\protect\sqrt{s}$ is set to be 500 GeV (a) and 1 TeV
(b). (From \protect\cite{hagi}.)}
\label{fig:Hagi}
\end{figure}


\section{Threshold behaviour}
The previous section dealt with top quark production sufficiently far
above threshold for the reaction to be well described by the Born
cross section, modified slightly by QCD and electroweak corrections.
This is in contrast to the situation in the threshold region, where
QCD plays an important role and controls the cross section.  Strong forces
modify the Born prediction. They compensate the phase space suppression and
enhance the production rate significantly, leading to a step function like
behaviour at threshold.  The large top decay rate also plays an important
role. Quarkonium resonances cease to exist and merge into a structureless
excitation curve which joins smoothly with the continuum prediction above
the nominal threshold.

This sharply rising cross section allows to study top quarks in a
particularly clean environment and with large rates. The following physics
questions can be addressed:
\begin{itemize}
\item The QCD potential can be scrutinized at short distances, with the
non perturbative tail cut off by the top decay. As a result
$\Lambda_{QCD}$ or $\alpha_s$ could be determined accurately.
\item The top quark mass can be measured with a precision which is only
limited by the theoretical understanding of the excitation curve, but in
any case better than 500 MeV.
\item Top quarks are strongly polarized (about 40\%) even for unpolarized
beams; and longitudinal beam polarisation will enhance this value even
further. Detailed studies of top decays, in particular of the $V-A$ structure
of the $tbW$ coupling are therefore feasible.
\item The interquark potential is --- slightly --- modified by the Yukawa
potential induced by Higgs exchange.  The excitation curve and the top
quark momentum distribution may therefore lead to an indirect measurement
of the Yukawa coupling.
\item The large number of top quarks in combination with the constrained
kinematics at threshold, could facilitate the search for new decay modes
expected in extensions of the SM.
 \end{itemize}
With this motivation in mind the following points will be discussed: After
a brief review of qualitative features of threshold production (section
3.2.1), the present status of our theoretical understanding of the total
cross section will be presented in section 3.2.2.

The momentum distribution of top quarks and their decay products offers an
alternative and complementary route to probe the interquark potential, as
shown in section 3.2.3.  Spin effects and angular distributions are
sensitive towards the small $P$-wave contribution induced by the axial part
of the neutral current.  The theoretical framework and the resulting
predictions are collected in section 3.2.4.  Rescattering, relativistic
corrections and other terms of order $\alpha_s^2$ will be touched upon in
section 3.2.5.

\subsection{Introductory remarks}

For a qualitative understanding it is illustrative to compare the different 
scales which govern top production close to threshold.  The quarks are 
produced at a scale comparable to their Compton wave length
\be
d_{\rm prod} \sim 1/m_t
\ee
Electroweak vertex corrections do not alter this behaviour significantly, 
since $Z$- or $W$-boson exchange proceeds at a distance $\approx 1/m_Z$ 
which is still short compared to scales characteristic for the bound state 
dynamics. For the QCD potential 
\be
V_{\rm QCD} = -{4\over 3}{\alpha_s\over r}
\ee
one anticipates an effective coupling constant ${4\over 3}\alpha_s \approx 
0.2$, if $\alpha_s$ is evaluated at the scale of the Bohr momentum
\be
k_B \approx    {4\over 3}\alpha_s {m_t\over 2} \approx 20 GeV
\ee
The resulting Bohr radius 
\be
r_{\rm Bohr}= 1/k_B
\ee
is small compared to hadronic scales. The binding energy of the 1S level 
\be
E_B =  \left( {4\over 3}\alpha_s\right)^2 {m_t\over 4}\approx 2 
GeV
\ee
and, quite generally, the separation between different resonances, is 
smaller than the decay rate 
\be
2\Gamma_t \approx 3 GeV
\ee
whence all resonances will merge and join smoothly with the continuum.  

The coupling strength $\kappa$ of the Yukawa potential 
\be
V_Y = -{\kappa\over r}e^{-m_Hr}\quad {\rm with } \quad 
\kappa = \sqrt{2}G_F {m_t^2\over 4\pi} = 0.042
\ee
is comparable to the QCD coupling $4/3\alpha_s=0.2$ in magnitude. The 
exponential damping, however, with a cutoff $1/m_H \ll r_{\rm Bohr}$ and a 
lower limit $m_H > 65 $ GeV, reduces the impact of the Yukawa potential 
quite drastically.  (The  situation may be different in multi-Higgs-models: 
the couplings could be enhanced and, even more important, the Higgs might be 
lighter!)  Furthermore, the nonrelativistic treatment is no longer adequate 
and retardation effects must be taken into consideration.

The large top quark width plays a crucial role for the threshold behaviour, 
which is best understood in the framework of (nonrelativistic) Green's 
function techniques. The production of $t\bar t$ from a pointlike source 
(actually of extension $1/m_t$) at $x'$ with frequency $\omega = E$ is 
characterized by the time dependent Green's function  $G(\vec r,\vec r',t)$ 
which is a solution of the time dependent Schr\"odinger equation with a 
pointlike source term
\be
(i\partial_t-H) G(\vec r,\vec r',t) =  \delta(\vec r - \vec r') e^{-iEt}
\ee
with
\be
H={p^2\over 2m} + V(\vec r)
\ee
In the problem at hand $m=m_t/2$ is the reduced mass, $\vec r$ the relative 
distance between $t$ and $\bar t$, and the width $\Gamma=2\Gamma_t$.  The 
location of the source is at the origin
 $\vec r'=0$ by convention, and the second argument 
of $G$  will be suppressed, $G(\vec r,t)\equiv G(\vec r,0,t)$.  For a 
qualitative discussion one may ignore the potential and obtains for a
stable quark
\be
G(\vec r,t)&=& -{m\over 2\pi r} e^{i\sqrt{2mE}r} e^{-iEt}
\nonumber\\
\widetilde G(\vec p,t) &=& {1\over E-{p^2\over 2m}} e^{-iEt}
\label{eq:Green}
\ee
The corresponding current is flowing in radial direction from the source
\be
\vec j = {i\over 2m} \left( G\vec\nabla G^* -G^*\vec\nabla G \right)
=\vec e_r {m^2v\over 4\pi^2r^2}
\ee
with a constant flux through a sphere around the origin, reflecting the 
conservation of probability.

The width $\Gamma = 2\Gamma_t$ is introduced in the Schr\"odinger equation 
through the replacement
\be
H\to H-i\Gamma/2
\ee
and, consequently, through the substitution $E\to E+i\Gamma/2$ in 
eq.~(\ref{eq:Green}). The exponential damping of the flux in radial 
direction
\be
\vec j= \vec e_r {m^2 v\over 4 \pi^2 r^2} e^{i(\sqrt{2m(E+i\Gamma/2)}
-\sqrt{2m(E-i\Gamma/2)})r}
\ee
is most easily interpreted in two limiting cases.  For $E\ll \Gamma$ the 
decrease 
\be
\vec j\sim  {\vec e_r\over r^2} e^{-\sqrt{2m\Gamma}r}
\ee
is solely driven by the large width, with a cutoff $x_c^{-1} = 
\sqrt{2m\Gamma} \approx 2.4$ GeV for realistic parameters.  For $E\gg 
\Gamma$, on the other hand, the current decreases like
\be
\vec j\sim  {\vec e_r\over r^2} e^{-\Gamma r/v}
\ee
In this latter case top quarks may travel appreciable distances, up to 
$x_{\rm cut} \approx v/\Gamma$. However, for realisitic beam energies
they hardly propagate beyond the perturbative region.

%\subsection{The total cross section}

\subsubsection{Predictions for  the Coulomb potential}
The large top decay rate restricts the range of sensitivity to the short 
distance part of the potential
\be
\widetilde V(q) = -{16\pi\over 3} {\alpha_V(q)\over q^2}
\ee
which is approximately Coulombic, with a logarithmic variation of 
$\alpha_V$. Most of the qualitative features of top quark threshold physics 
can be understood even on the basis of the results for constant $\alpha_V$, 
which are available in analytical form. 

A remarkable feature of heavy quark production is the sharp rise of the 
cross section at threshold, a consequence of the attractive Coulomb force.  
The step function joins smoothly with the smeared resonances. 

Let us try to quantify this aspect with the help of simple nonrelativistic 
quantum mechanics ($\alpha={4\over 3}\alpha_s$, $m=m_t/2$).  The narrow 
resonances below the nominal threshold ($E$=0) are located at $E_n = 
-E_{Ryd}/n^2$ with $E_{Ryd}=k_B^2/(2m)=\alpha^2m/2$;

The production amplitude from a pointlike source with frequency $\omega=E$ 
is proportional to
\be
\langle \vec r=0 | \psi_n\rangle = \psi_n(0),
\ee
the rate correspondingly to
\be
R\sim |\psi_n(0)|^2\delta(E-E_n)
\ee
The wave function at the origin decreases with the third power of the 
radial quantum number,
\be
 |\psi_n(0)|^2 \sim 1/n^3
\ee
their spacing becomes increasingly dense
\be
\Delta E = E_{n+1}-E_n = E_{Ryd} \left({1\over n^2} -  {1\over 
(n+1)^2} \right)\sim E_{Ryd} {2\over n^3}
\ee
such that their average contribution to the cross section approaches a 
constant value (Fig.~\ref{fig:S18}).
  Above threshold one has to project the state $\langle \vec 
r=0|$ onto the Coulomb wave functions in the continuum $\psi_E(\vec r)$.
These replace the plane waves which are appropriate for the case where
final state interaction is absent.

The production amplitude is thus governed by
\be
\langle\vec r=0| \psi_E\rangle = \psi_E(0)
\ee
and the rate $\sim |\psi_E(0)|^2$. The threshold phase space factor $v$ 
is thus compensated by the $1/v$ singularity in $ |\psi_E(0)|^2$, and the 
cross section approaches a constant value for $E\to 0$. 
\begin{figure}[htb]
%\psfig{figure=resonance.ps,width=100mm,bbllx=70pt,bblly=135pt,%
%bburx=515pt,bbury=550pt}
\psfig{figure=resonance.ps,width=120mm,bbllx=0pt,bblly=270pt,%
bburx=450pt,bbury=550pt}
    \caption{Schematic representation of resonances (wiht the area of
the boxes adjusted to represent the weights of the respective delta
functions) and the continuum cross section.}
\label{fig:S18}
\end{figure}

The explicit calculation yields 
\be 
R(E) = {9\pi\over 2\alpha^2} \Gamma_e \delta(E-E_n) = 3Q_t^2{3\over 2} 
\sum_n {k_B^2\over m^2} {1\over n^3} \pi \delta\left(E-E_{Ryd}{1\over 
n^2}\right)
\label{eq:RE}
\ee
   for energies below threshold, and
\be
R(E)=3Q^2_t {3\over 2} \beta {x\over 1-e^{-x}} 
\label{eq:347}
\ee
with $x=k_B/k = \pi\alpha/\beta$
for energies above threshold.\footnote{For a textbook discussion of Coulomb 
scattering states and a derivation of this result see 
e.g.~\cite{Landau,Messiah}}

The perturbative expansion breaks down in the limit $\beta\to 0$.  The 
first term of this formal series
\be
\beta   {x\over 1-e^{-x}}  = \beta \left( {x\over 2}-\ldots\right)
\ee
underestimates the exact result by a factor two.



Eq.~(\ref{eq:RE}) allows to connect the formalism based on narrow individual 
resonances with a formulation which is tailored to the situation at hand, 
namely wide overlapping resonances which merge into a smooth continuum.


Instead of summing 
the contributions from a large number of high radial excitations
one may directly calculate the imaginary part of the Greens
function for complex energy
\begin{eqnarray}
\sigma (e^+e^- \to t\bar{t})
      = \frac{24\pi^2\alpha^2}{s} \frac{\rho_v(s)}{m^2_t}
       \left( 1 - \frac{16}{3} \frac{\alpha_s}{\pi} \right)
       \sum_{n} \left| \psi_n (0) \right|^2
       \frac{\Gamma_t}{(E_n - E)^2 + \Gamma^2_t} \nonumber\\
      = \frac{24\pi^2\alpha^2}{s} \frac{\rho_v(s)}{m^2_t}
       \left( 1 - \frac{16}{3} \frac{\alpha_s}{\pi} \right)
       \sum_{n} Im \frac{\psi^{\*}_n (0) \psi^*_n (0)}
       {E_n - E - i \Gamma_t} \nonumber\\
      = -\frac{24\pi^2\alpha^2}{s} \frac{\rho_v(s)}{m^2_t}
       \left( 1 - \frac{16}{3} \frac{\alpha_s}{\pi} \right)
       Im G(0,0,E+i\Gamma)
\label{eq:60}
\end{eqnarray}
The factor  $\rho_v(s)$  incorporates the contributions from the
intermediate photon and $Z$ and is given by 
\begin{eqnarray}
\rho^{Born}_v(s) = \left| e_t e_e + \frac{1}{y^2}
                \frac{\upsilon_t \upsilon_e M^2_\theta}
                     {s - M^2_Z + iM_Z \Gamma_Z} \right|^2
              + \left| \frac{1}{y^2}
                \frac{\upsilon_t a_e M^2_G}
                     {s - M^2_Z + iM_Z \Gamma_Z} \right|
\label{eq:58}
\end{eqnarray}
\[ \upsilon_f = 2I_{3f} - 4e_f \qquad
   a_f = 2I_{3f} \qquad
   y = 16 \sin^2 \theta_W \cos^2\theta_W \]
($\alpha=\alpha_{\rm eff}=1/128$ has been adopted in the numerical
   evaluation. Radiative corrections to this formula have been discussed in
\cite{GKKS}.)

The problem can be solved in closed analytical form for an exact
Coulomb potential \cite{fk} 
\be
{\rm Im}G_{E+i\Gamma_t} (0,0) &=& -{m_t^2\over 4\pi}
\left[ {k_2\over m_t} + {2k_B\over m_t}\arctan {k_2\over k_1}
\right.
\nonumber \\
&&\left. 
 +\sum_{n=1}^\infty {2k_B^2 \over m_t^2\pi^4} 
{\Gamma_t k_B n + k_2 \left(n^2
\sqrt{E^2+\Gamma_t^2}+{k_B^2\over m_t}\right)
\over
\left( E+{k_B^2\over m_tn^2}\right)^2+\Gamma_t^2}\right]
\nonumber\\
k_{1,2} &=& \left[
m_t\left( \sqrt{E^2+\Gamma_t^2}\mp E\right)/2\right]^{1/2},
\nonumber\\
k_B&=& {2\over 3} \alpha_s m_t.
\ee



To arrive at a realistic prediction of the total (and, in sect 3.2.3
of the differential) cross section the Coulomb potential must be
replaced by the realistic QCD potential.



\subsection{The QCD potential}

On the basis of earlier conceptual work in \cite{apple,sus} the
asymptotic behaviour of the static potential has been derived in
\cite{fis,bil}.  In momentum space the potential reads in the
$\overline{MS}$ subtraction scheme
\begin{equation}
V\left( Q^2 , \alpha _{\overline{MS}} (Q^2)\right) = - \frac{16\pi}{3}
 \frac{\alpha _{\overline{MS}} (Q^2)}{Q^2}
  \left[ 1 + \left( \frac{31}{3} -
  \frac{10}{9} n_f \right)
   \frac{\alpha _{\overline{MS}}(Q^2)}{4\pi} \right]
\label{eq:49}
\end{equation}
The renormalization scale $\mu ^2$ has been chosen as $Q^2$, and $n_f$
refers to the number of massless quarks.  Employing standard arguments based
on the renormalization group, the $Q^2$ expansion of $\alpha _{\overline{MS}}(Q^2)$
is given by
% eq typed by chetyrkin
\begin{eqnarray}
\frac{\alpha_{\overline{MS}}(Q^2)}{4\pi } &=&
\frac{1}{b_0 \log
 \left( Q^2/\Lambda^2_{\overline{MS}}
 \right)}
 \left[    1 - \frac{b_1}{b^2_0}
          \frac   {\log\log
          \left(  Q^2 / \Lambda^2_{\overline{MS}} \right) }
                  {\log\left( Q^2 / \Lambda^2_{\overline{MS}}
          \right) }
 \right]\quad
\label{eq:50}\\
b_0 &=& 11 - \frac{2}{3}n_f, \qquad
b_1 = 102 - \frac{38}{3}n_f
\nonumber
\end{eqnarray}

The leading behaviour of the potential at small distances
$[\gamma_E=0.5772\cdots]$
\begin{eqnarray}
V(r) & = & \frac{16\pi}{(33-2n_f)r \log 1 / (\Lambda _{\overline{MS}}r)^2}
\nonumber \\
& & \left[ 1 - \frac{b_1}{b^2_0}
               \frac{\log\log 1 / (\Lambda _{\overline{MS}}r)^2}
               {\log 1 / (\Lambda _{\overline{MS}}r)^2}
             + \frac{\left( \frac{31}{3} - \frac{10}{9} n_f \right)
               / b_0 + 2 \gamma_E }
               {\log 1 / (\Lambda _{\overline{MS}}r)^2 } + \cdots \right]
\label{eq:51}
\end{eqnarray}
is thus directly given in terms of the QCD scale parameter $\Lambda$.
The exploration of $V(r)$ for small distances could thus lead to a
direct determination of $\Lambda$.  For quark masses above 50-100~GeV the
ground state properties become independent of the potential in the
nonperturbative region.  As discussed in the previous section the
large decay rate acts as a cutoff and the predictions are fairly
insensitive to the actual regularisation.  However, an additional
constant which can be traded against a shift in $m_t$ must be
carefully calibrated.

In practice one connects the theoretically predicted short distance part
smoothly with the empirically determined potential above $\sim$0.1~fermi.
The asymptotic form given in (\ref{eq:49})
is based on the assumption that $n_f$ species of light quarks, taken
as massless, contribute to the vacuum polarization, and heavier ones
are ignored.  The value of $\Lambda_{\overline{MS}}$ in (\ref{eq:50})
must be properly related to $\Lambda_{\overline{MS}}$ as determined
from other experiments with a different number of effective light
flavors \cite{bern,marc}.  For the momentum range of around 15~GeV explored
by the $t\bar{t}$ system $n_f=$5 seems adequate. 

In the subsequent discussion the Green's function will be calculated in 
momentum space with the help of the Lippman-Schwinger integral
equation \cite{JKT}.  The 
representation of the QCD potential in momentum space with the large $Q^2$ 
behaviour given by eqs.~\ref{eq:49} and \ref{eq:50} will be employed.
The intermediate  
and small momentum dependence will be based on Richardson's potential. This 
choice allows to vary $\alpha_{\overline{MS}}(M_Z^2)$ (or equivalently 
$\Lambda_{\overline{MS}}$) between 0.11 and 0.13, while maintaining a 
smooth $Q^2$ dependence of $\alpha_V(Q^2)$ (Fig.~\ref{fig:AEFF}).
\begin{figure}[htb]
  \begin{center}
    \leavevmode
    \epsfxsize=60mm
    \epsffile[130 600 390 610]{fig_3.12.ps}
\vspace*{110mm}
    \hfill
\caption{$\alpha_{\rm eff}$ for different values of $\alpha_s(M_Z)$:
solid: 0.12, dashed: 0.11, dashed-dotted: 0.13, dotted: 0.10 and 0.14.}
\label{fig:AEFF}
  \end{center}
\end{figure}
  An additive constant 
in coordinate space (corresponding to a $\delta$-function in momentum 
space)  is adjusted to fix $V(r=1 GeV^{-1})=-1/4$ GeV for arbitrary  
$\alpha_{\overline{MS}}$. This constraint avoids the unmotivated and 
uncontroled variation of the long distance part of $V(r)$ with a change in 
$\alpha_{\overline{MS}}$. The potential in coordinate space is shown in 
Fig.~\ref{fig:COOR}.

\begin{figure}[ht]
  \begin{center}
    \leavevmode
    \epsfxsize=12.cm
    \epsffile[70 135 515 710]{fig_3.13.ps}
    \hfill
\caption{QCD potential in the position space $V_{JKT}$
 for different values of $\alpha_s(M_Z)$:
solid: 0.12, dashed: 0.11, dashed-dotted: 0.13, dotted: 0.10 and 0.14.}
\label{fig:COOR}
  \end{center}
\end{figure}

%\begin{figure}[htb]
%%\vspace*{100mm}
%\psfig{figure=fig_3.13.ps,width=120mm,bbllx=70pt,bblly=135pt,%
%bburx=515pt,bbury=710pt}
%\caption{QCD potential in the position space $V_{JKT}$
% for different values of $\alpha_s(M_Z)$:
%solid: 0.12, dashed: 0.11, dashed-dotted: 0.13, dotted: 0.10 and 0.14.}
%\label{fig:COOR}
%\end{figure}
